Author’s response to reviews

Title: Impact of a probiotic fermented milk in the gut ecosystem and in the systemic immunity using a non-severe protein-energy-malnutrition model in mice

Authors:

Carolina Maldonado Galdeano (cmaldo@cerela.org.ar)
Ivanna Novotny Nuñez (inovotny@cerela.org.ar)
Alejandra de Moreno de LeBlanc (demoreno@cerela.org.ar)
Esteban Carmuega (esteban@carmuega.com.ar)
Ricardo Weill (ricardo.weill@danone.com)
Gabriela Perdigon (perdigon@cerela.org.ar)

Version: 3 Date: 17 March 2011

Author’s response to reviews:

San Miguel de Tucumán, March 17th, 2011

To whom it may concern,

As corresponding author, I’m sending the revised version of the manuscript entitled “Impact of a probiotic fermented milk in the gut ecosystem and in the systemic immunity using a non-severe protein-energy-malnutrition model in mice” for publication in BMC Gastroenterology by Maldonado Galdeano C, Novotny Núñez I, de Moreno de LeBlanc A, Carmuega E, Weill R and Perdigón G.

Considering that the Editor solicited the revision of the English, a native English speaking colleague helped to copyedit the paper and the abstract. All the changes are also highlight (with ‘tracked changes’) in the new version of the manuscript.

We hope that the present version of the manuscript fulfill your requirements.

Sincerely,

Gabriela Perdigon

Gabriela Perdigon
CERELA-CONICET
Chacabuco 145 San Miguel de Tucuman
Tucuman, Argentina C.P. 4000
Tel: 54-381-4310465 Ext 129
Fax: 54-381-4005600
perdigon@cerela.org.ar